Author’s response to reviews

Title: Electroacupuncture at LI11 Promotes Jejunal Motility via the Parasympathetic Pathway

Authors:

Xuanming Hu (drhxm@qq.com)
Mengqian Yuan (515347441@qq.com)
Yin Yin (837297259@qq.com)
Yidan Wang (2937865782@qq.com)
Yuqin Li (610356657@qq.com)
Na Zhang (965298484@qq.com)
Xueyi Sun (1304758284@qq.com)
Zhi Yu (mickey28282@sina.com)
Bin Xu (xuuuux@sina.com)

Version: 4 Date: 03 Jun 2017

Author’s response to reviews:

Dear editor:

We appreciate sincerely the comments and suggestions made by you and the reviewers on our manuscript (Electroacupuncture at LI11 promotes jejunal motility via the parasympathetic pathway). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have carefully revised the manuscript accordingly and hope you would find this version acceptable for publication.

We have included the email addresses of all manuscript authors on our Title Page, and also altered our supplementary files to a section within all of the information.
Thank you very much for your favorable consideration of this manuscript.

Best regards,

Xuanming HU